The Day After Tomorrow: Evaluating the Burden of Trump's Trade War^{*}

Meixin Guo

School of Economics and Management Tsinghua University Beijing, China guomx@sem.tsinghua.edu.cn

Lin Lu

School of Economics and Management Tsinghua University Beijing, China Iulin@sem.tsinghua.edu.cn

Liugang Sheng

Department of Economics The Chinese University of Hong Kong Shatin, N.T., Hong Kong Isheng@cuhk.edu.hk

Miaojie Yu

National School of Development CCER Peking University Beijing, China mjyu@nsd.pku.edu.cn

Abstract

During his U.S. presidential campaign Donald Trump threatened China with the imposition of high import tariffs on its exports to the United States. To evaluate the repercussions of such an action, this paper uses Eaton and Kortum's 2002 multi-sector; multi-country general equilibrium model with intersectional linkages to forecast how exports, imports, output, and real wages would change if Trump's threat of 45 percent tariffs is carried out. To view plausible scenarios, we evaluate the case of a unilateral action on the part of the United States, as well as a scenario where China retaliates by imposing an equally high 45 percent tariff on its imports from the United States. In addition, because the high U.S. trade deficit with China is a factor that underpins calls for tariff action, we explore simulations where the trade balance is restored to balance as well as a scenario in which the trade balance is unchanged. In all of the scenarios, the calibration exercise suggests that a trade war triggered by high U.S. import tariffs will lead to a collapse in U.S.–China bilateral trade. In all of the scenarios, the United States will experience large social welfare losses, whereas China may lose or gain slightly depending on the effect of trade war on the U.S.–China trade balance. Globally, some small open economies may experience small benefits, while other countries may suffer collateral damage.

^{*} We thank Professors Wing Thye Woo, Fukunari Kimura, Barry Eichengreen, and participants of the Asian Economic Panel Meeting in Kuala Lumpur in March 2017 for their helpful suggestions and comments. We thank Gu Yan, Kai Mu, and Yue Zhou for their excellent research assistance. We also thank the National Science Foundation of China grants 71203121 (project organized by Meixin Guo) and 71303128 (project organized by Lin Lu), as well as Humanities and Social Sciences Research Fund Supported by Tsinghua University grant 2013WKYB004 (project organized by Lin Lu) for financial support. Finally, we thank the National Science Foundation of China for grants 71203121 (Meixin Guo), 71303128 (Lin Lu), 71625007, and 71573006 (Miaojie Yu). All remaining errors are our own.

I. Introduction

During his bid for the U.S. presidency, Donald Trump injected strong calls for protectionism. Following his election, Trump used his inaugural speech to call for "America First" and for "Buy American, Hire American." He also started to execute his campaign pledges breaking the trade ties of the United States with its neighboring countries and main trade partners. For instance, Trump formally withdrew the United States from the Trans-Pacific Partnership, an agreement among 12 countries across three continents that took nearly ten years to negotiate under his predecessor, the former U.S. President Barack Obama. He also signed an executive order to build a wall along the Mexican border and threatened Mexico to pay for its construction by increasing taxes on its exports to the United States. He ordered his team to initiate a renegotiation of NAFTA between the United States, Mexico, and Canada. These actions, along with many others, have dispelled any remaining doubt over the sincerity of President Trump's promises during the election campaign. During the recent meeting of G20 finance ministers and central bankers, the financial leaders of the world's biggest economies dropped a pledge to keep the global trade free and open, thereby acquiescing to the increasingly protectionist United States.

In light of the administration's protectionist actions, it is important to ask whether the United States will pull the trigger on a trade war against the country's main trade partners, such as China. China has been among the primary targets of President Trump's campaign and administration. In his Monessen, Pennsylvania, speech on 28 June 2016, Trump condemned China's entry to the World Trade Organization (WTO) as a catastrophe for U.S. manufacturing workers. Later, when he met with the Editorial Board of the *New York Times* in January 2016 he proposed the idea of imposing 45 percent import tariffs on China's exports to the United States. In one of his well-known tweets, President Trump also described China as the "grand champion" of currency manipulation to boost her exports. Because it is no longer implausible to imagine the outbreak of a trade war between the United States and China, it is important to evaluate how such a conflict would affect the United States and China, as well as other countries.

To forecast the economic implications of a trade war, we adopt Eaton and Kortum's (2002) multi-country and multi-sector general equilibrium model, modified to include intersectoral linkages of the form proposed by Caliendo and Parro (2015). Through this exercise, we are able to examine the changes in the exports, imports, output, and real wages of 62 major economies in response to a hypothetical 45 percent tariff levied by the United States on its imports from China. We consider three possible tariff hike scenarios involving agriculture, mining, and manufacturing sectors. In the first scenario, the United States increases its import tariffs to 45 percent on all imports from China and all countries have balanced trade. Balanced trade might well be one of the goals of a trade war as the U.S. government has for a long time blamed China for its large trade surplus.¹ In the second case, we assume China retaliates by also increasing its tariffs on its U.S. imports to 45 percent, again evaluating the impacts under the assumption of balanced trade. In the third case, we consider a situation in which both China and the United States impose high tariffs to each other, though trade imbalances for all countries remain unchanged. For simplicity, we name these three cases as (1) unilateral U.S. tariffs with balanced trade, (2) U.S.–China retaliatory tariff war with balanced trade, and (3) U.S.–China retaliatory tariff war with ongoing trade imbalance, respectively.

Our exercise shows that in all scenarios, high U.S. import tariffs lead to a catastrophic collapse in bilateral trade. In the case of unilateral U.S. tariffs with balanced trade, for half of China's 18 tradable sectors, exports to the United States fall by more than 90 percent. These sectors include textiles, metal products, computers, and electrical equipment. In the next two cases, the trade war between China and the United States leads to a similar collapse in bilateral trade. In these scenarios the dramatic decline in trade includes agriculture, mining, and petroleum products, in addition to computers and electrical equipment.

The consequence of a trade war that substantially impairs bilateral trade will be a slump in output and significant decline in social welfare. In the case where both countries engage in trade war-that is, U.S.-China retaliatory tariff war with balanced trade-China's output in textile and computer products is predicted to fall by 6.29 percent and 14.26 percent, respectively. At the same time, the output declines in U.S. agriculture and food industries will amount to 1.14 percent and 4.18 percent, respectively. To measure social welfare loss, we use changes in real wages before and after the trade war, which accounts for the effects of a rising price index. In the first two scenarios, the United States suffers large losses while China only bears a small welfare loss. By our calculation, the United States experiences welfare losses of 0.66 percent and 0.75 percent, respectively, compared with China's maximum loss of 0.04 percent in the case of unilateral U.S. tariffs with balanced trade. In the third case, with unbalanced trade, China experiences the largest losses (-0.37 percent)and the losses of the United States are the second largest (-0.32 percent). Some other countries in Asia may gain slightly from the diversion, whereas other advanced economies may experience collateral damage due to spillover effects transmitted through input-output linkages and the general equilibrium effects of the trade war between the two largest global economies.

Our study highlights the important role of the trade imbalance in the evaluation of the consequences of a China–U.S. trade war. The trade imbalance matters because one country may finance its consumption through trade deficits when their labor and tariff revenue

¹ For simplicity, we assume all countries achieve trade balance after the trade war starts. We will also consider the case in which trade imbalance remains.

income is lower in value than their total expenditure. Thus, a country with a trade deficit receives a net income transfer from other countries. Because we use a static trade model to simulate the effects of tariff changes on output and trade, the trade balance is exogenously determined by assumptions. For simplicity, we consider two possibilities: the trade war restores trade balance or maintains the current trade imbalance. The reality would be somewhere in between, but these two possible scenarios explain that the United States is put in a disadvantage because of current large trade deficits, and China may gain slightly from the trade war if trade achieves rebalance.

Given the current U.S.-China trade imbalance, the presumption of whether the trade war leads to trade balance or not has different implications for the two countries. If the trade war leads to a trade balance, the United States must export more and import less to move from a trade deficit to balanced trade, whereas China needs to reduce its exports more than it reduces its imports to achieve balance. In other words, the United States will not receive the net income transfer, which acts a negative income shock. This shock may be further increased if China hikes its tariffs. In contrast, because China will not need to pay for the net income transfer, the reduction in its trade surplus with the United States may alleviate the negative effect of trade war on its economy. Thus, in this case social welfare loss will be even larger for the United States, whereas China may even gain slightly if the positive income effect dominates, as shown in the U.S. vs. China with balanced trade case. In contrast, if the trade war does not have any effect on the current trade imbalance, China will effectively continue to pay the income transfer while the United States maintains its external borrowing, even the trade war reduces overall trade. Thus, China may be hurt more than the United States by the trade war, as shown in the case of U.S. vs. China with imbalanced trade. Our analysis implies that trade balance might not be a desirable target for the United States if it launches a trade war against China while planning to maximize U.S. social welfare.

Admittedly, the quantitative effects of Trump's trade war on output and social welfare are less striking than those on bilateral trade. Nevertheless, our calculation of welfare loss is rather conservative and likely to underestimate the effect of the possible trade war on output and social welfare. One key assumption in our model is that all economies function well without any other frictions, aside from trade costs. Given that labor is freely mobile across all sectors within each country, the sectoral reallocation between tradable and non-tradable sectors, together with the import substitution among different source countries, can offset the effects of unilateral import tariff hikes imposed by the other country. Moreover, the presence of input–output linkages also diminishes the effects of unilateral tariff hikes. In reality, however, these adjustments may not be smooth, in which case the impact of a trade war on the world economy will be magnified. Finally, though a trade war is likely to trigger a tsunami in global financial markets, these effects are not accounted for in our framework.

One of the most popular approaches for evaluating the possible consequences of a trade war is the traditional computational general equilibrium (CGE) model, which fully specifies a parametric model of preferences, technology, and trade costs with ad hoc parameters. Our approach differs from this model by following the recent development in quantitative trade models, largely triggered by the seminal work of Eaton and Kortum (2002). The extension of the Eaton and Kortum (EK) model into a multi-sector model with input–output linkages and other features has become the workhorse model for counterfactual analysis. This approach is suitable for analyzing trade policy changes and offers at least three significant advantages over traditional CGE models or the recently developed CGE model with Melitz-type firm heterogeneity (Petri, Plummer, and Zhai 2012) for the following reasons.

First, the EK model offers more parsimony by including a limited number of parameters. The latest version of the Global Trade Analysis Project model has approximately 13,000 parameters that cannot be estimated, whereas researchers who adopt new quantitative trade models generally use data to estimate the key parameters before conducting counterfactual analysis. Second, the new quantitative trade models have appealing microtheoretical foundations. For example, one does not need to assume that each country produces one distinct good—the so-called Armington assumption—to do quantitative work in international trade. Third, although the CGE model combined with Melitz can capture firm heterogeneity, it is not only difficult to generate a sectoral gravity equation with macro implications but also intractable to identify a rich set of related fixed costs using the actual data. By contrast, the EK model can deliver gravity equations that even incorporate a country's trade deficit/surplus.

Many recent studies have applied or extended the EK framework to analyze various topics, including the evaluation of the possible gains from a trade agreement, technological changes, and infrastructure improvement. For example, Donaldson (2010) applied the EK model to empirical data and assessed the gains from railroad construction in colonial India. Caliendo and Parro (2015) extended EK framework to include input-output linkage and evaluated the gains from NAFTA.² Dekle, Eaton, and Kortum (2008) showed that the EK framework can also be used to analyze hypothetical cases, such as how much the U.S. GDP needs to adjust to eliminate its high current account deficits. The rapid development in this approach provides suitable tools for us to evaluate the possible outcomes of a trade war triggered by the largest economy in the world.

² Di Giovanni, Levchenko, and Zhang (2014) adopted a similar framework to evaluate the gain from China's trade integration with the world market and its fast technological changes. A few recent studies have introduced labor migration into the EK framework and explored the impact of goods and labor market frictions on economic growth and gain from trade (Galle, Rodriguez-Clare, and Yi 2015; Caliendo and Parro 2015; Tombe and Zhu 2015).

The remainder of this paper is organized as follows: Section 2 reviews the bilateral trade relationship between the United States and China, the dynamics of the bilateral trade, and the current trade conflicts. Section 3 presents our model, data, and calibration method. Section 4 shows the calibration results. Section 5 presents the concluding remarks with a discussion on trade policies.

2. Overview of the trade relationship between the United States and China

2.1 Bilateral trade relationship

At the establishment of the People's Republic of China (PRC, or China) in 1949, the United States retained its diplomatic recognition of Taipei instead of Beijing. The diplomatic and economic interactions between the United States and China were at their lowest level during the following years of the Cold War, since conflicts in ideology and national security interests greatly impeded bilateral trade between these nations.

Following the China–Soviet border conflicts in the late 1960s, both China and the United States began to realize the potential benefits of normalizing a bilateral relationship. In June 1971, U.S. President Nixon ended the legal barriers of trade with China, and his ice-breaking visit to China in 1972 further provided a pathway to the thawing of trade relations between the two countries.

Following China's 1978 market-oriented economic reforms, the United States started to grant China Most Favored Nation (MFN) tariffs in January 1980, subject to annual renewal.³ MFN status implies that a country provides another partner with the same tariff treatment as it does under obligation to other countries who are formal members of the WTO, or its predecessor agreement, the General Agreements on Trade and Tariffs (GATT). MFN status significantly liberalized China's access to the U.S. market, because it supplanted the otherwise high tariffs that were levied on non-WTO countries. The United States soon became the second largest importer for China and China's third largest partner in 1986. In the same year, China applied for GATT membership and the United States was also interested in China's further trade and FDI liberalization. Thus, the annual waiver of the Jackson-Vanik Amendment and the congressional renewal of China's MFN status came to an end in 1999, and the United States granted China Permanent Normal Trade Relations, thereby paving the road for China to join the WTO in 2001.

The decade and a half following China's accession to the WTO has been a honeymoon for the two countries, and their bilateral trade has grown much faster than before. The United

³ The Jackson-Vanik Amendment of 1974, which denied preferential trade policies to some countries, was often targeted at communist countries. Although the application of this amendment was waived by U.S. presidents, the amendment required an annual congressional renewal of China's MFN status.

States and China have become each other's most important trade partner, though trade conflicts have continued. For instance, China's large trade surplus and inflexible exchange rate have been criticized frequently by the U.S. government. The United States has also accused China of dumping textile, steel, and other manufactured products at unfairly low prices. The George W. Bush and Obama administrations imposed quotas and high tariffs on the imports of Chinese textile and other low-end industrial products to protect U.S. domestic industries. These trade conflicts have not reversed the movement toward freer trade, however. Thus, the 2017 start of the Trump administration brings concerns for the prospects of further liberalization because of Trump's open support of protectionism.

2.2 Bilateral trade flow and trade imbalance

We examine three aspects of China–U.S. trade, namely, bilateral trade flows and trade imbalance, bilateral trade structure and trade dispute in key industries such as steel, and current trade conflicts.

The trade volume between China and the United States has grown rapidly over the last three decades, especially following China's accession to the WTO in 2001. The bilateral trade volume has surged from US\$ 97 billion in December 2001 to more than US\$ 524 billion in 2016, which implies an average annual growth rate of 11.11 percent.

The annual growth of bilateral trade volume between these two countries has slowed since 2008 partly because of the 2008–09 global financial crisis. The China–U.S. trade volume also shrank by 6.26 percent in 2016—the first instance of negative growth since 2009. Although exports edged down by 5.13 percent in 2016, imports decreased by 9.79 percent following a 5.91 percent decline in 2015.

The fast-growing trade volume between the United States and China has been accompanied by a persistent bilateral trade surplus in China's favor. As shown in Table 1, China's trade surplus reached US\$ 260.37 billion in 2015 from only US\$ 30 billion in 2000. This unbalanced trade eventually resulted in a long-lasting dispute in China–U.S. relations. As the bilateral trade volume growth slowed down recently, the trade surplus growth also started to cool down. China's bilateral trade surplus narrowed by 2.45 percent to US\$ 253.99 billion in 2016, thereby reflecting a tendency toward a more balanced bilateral trade structure.

2.3 Bilateral trade structure and trade dispute

Table 2 presents bilateral trade flows in three main sectors: steel, textiles, and machines and computers. Machines and computers are China's leading exports to the United States, accounting for 44.45 percent (US\$ 173 billion) of its total exports in 2016. These products are followed by textiles, which represented 11 percent (US\$ 42.42 billion) of China's exports to the United States. China's exports in traditional competitive industries, however, shrank

	Trade flows,	billion USD	Growth rate, percent		
Year	Musa,chn	M _{CHN} , usa	Musa, CHN	M _{CHN} , usa	
2000	52.14	22.36			
2001	54.32	26.20	4.17	17.17	
2002	69.96	27.23	28.79	3.91	
2003	92.51	33.88	32.23	24.44	
2004	124.97	44.65	35.09	31.78	
2005	162.94	48.73	30.38	9.14	
2006	203.52	59.22	24.90	21.52	
2007	232.76	69.86	14.37	17.96	
2008	252.33	81.50	8.41	16.66	
2009	220.90	77.46	-12.45	-4.95	
2010	283.37	102.06	28.28	31.76	
2011	324.56	122.14	14.54	19.68	
2012	352.00	132.88	8.45	8.79	
2013	368.48	152.55	4.68	14.81	
2014	396.15	159.19	7.51	4.35	
2015	410.15	149.78	3.53	-5.91	
2016	389.11	135.12	-5.13	-9.79	

Table 1. Bilateral trade between the United States and China

Source: CEIC.

Note: M_{USA,CHN} denotes the total imports of the United States from China.

 $M_{USA,CHN} + M_{CHN,USA}$ denotes the total trade volume. $M_{USA,CHN} - M_{CHN,USA}$ denotes China's trade balance.

	Steel		Textiles		Machines a	nd computers
year	$M^{j}_{USA,CHN}$	M ^j _{CHN,USA}	$M_{USA,CHN}^{j}$	M ^j _{CHN,USA}	$M_{USA,CHN}^{j}$	$M^{j}_{CHN,USA}$
1993			3.31	0.23	2.93	3.84
1994			3.16	0.86	4.60	4.53
1995			3.17	1.35	5.53	5.13
1996			3.23	1.13	6.52	5.59
1997			3.57	0.99	8.34	5.37
1998			3.80	0.42	10.48	6.54
1999			3.98	0.24	12.48	8.02
2000			4.56	0.31	16.39	9.20
2001			4.57	0.35	17.99	11.38
2002			5.43	0.44	26.24	11.17
2003			7.19	1.08	39.39	11.42
2004			9.06	2.31	56.68	15.46
2005			16.67	2.11	72.79	16.84
2006			19.87	3.00	92.55	21.38
2007			22.90	2.42	107.85	23.72
2008	6.92	1.22	23.28	2.60	113.48	26.17
2009	1.51	0.90	24.60	1.71	104.72	22.32
2010	1.63	0.63	31.45	3.06	132.90	28.74
2011	2.58	0.65	35.06	4.18	150.01	29.45
2012	2.88	0.57	36.18	4.96	163.37	28.96
2013	2.75	0.58	38.95	3.82	169.34	38.31
2014	4.02	0.69	41.88	2.53	182.86	38.30
2015	2.85	0.58	44.79	1.98	179.89	35.67
2016	1.71	0.45	42.42	1.28	172.87	31.26

Table 2. Bilateral trade flows on selected sectors (billion USD)

Source: CEIC.

Note: $M_{j,SA,CHN}^{j}$ denotes the imports of the United States from China in sector *j*. The statistics of steel has been available since 2008, as the product "steel" was classified in the group of "iron and steels" before 2008.

in recent years, in accordance with the slowing pace of bilateral trade. Specifically, China's exports of machinery and computers, as well as textile products, decreased by 3.89 percent and 5.35 percent in 2016, respectively.

In terms of China's imports from the United States, machines and computers were also in first place, accounting for 23.13 percent (US\$ 31.26 billion) of China's total imports in 2016.⁴ This proportion reflects the intra-industry trade and the global production integration between these two countries. Therefore, a trade war is likely to have adverse effects on related industries.

Steel products have provoked ongoing and tense trade arguments between the United States and China. The United States has criticized China's official support of steel and aluminum products, claiming that China has distorted the global markets and dumped 100 million tons of steel into the global market. Between 2011 and 2015 the United States filed 29 anti-dumping and 25 anti-subsidy investigations against Chinese companies, of which 11 anti-dumping and 10 anti-subsidy investigations were concentrated on the steel industry.

2.4 Current trade conflicts

In the past two decades, and especially following China's WTO accession in 2001, both the United States and China realized significant gains from their trade liberalization and expanding bilateral markets. After President Trump's inauguration, however, the potential for trade dispute has intensified.

First, the U.S. government blamed its long period of slow GDP growth, weak employment growth, and sharp net loss of manufacturing employment on China's accession to the WTO. The U.S. government also argued that multilateral trade agreements (e.g., WTO rules) should be limited to countries that pursue free-market principles and implement transparent and functional legal and regulatory systems.

Second, the United States has criticized China for its unequal treatment of foreign companies with measures in favor of domestic firms and state-owned enterprises (SOEs), including: (i) state-driven industrial policies that groom domestic firms, particularly favoring SOEs; (ii) a government procurement process that is biased toward domestic firms, such as the "secure and controllable" policy for information and communication technology; and (iii) techno-nationalism under the auspices of the "Made in China 2025" initiative.

In response to these criticisms, China has denied adopting the "secure and controllable" policies to limit foreign trade and notified the WTO Technological Barriers to Trade committee. In the case of the "Made in China 2025" initiative, the Chinese government

⁴ The proportion of machine and computer imports also dropped in recent years from 25.11 percent in 2013 to 23.13 percent in 2016.

promised to provide equal opportunities to foreign and domestic enterprises to strengthen the role of the market.

Third, the United States named China as a significant market barrier for their exporting firms. Specifically, the United States alleged that China has imposed export restraints (e.g., quotas and licensing) to benefit domestic downstream firms at the expense of foreign competitors. The United States also accused China of using anti-monopoly law investigations to protect its domestic industries.

Fourth, intellectual property rights have become a hot topic in recent years. The United States complained that its enterprises are often required to transfer their technology as a condition to secure investment approvals. The United States also criticized the poor protection and enforcement of trade secrets by the Chinese government.

3. Data

We follow Caliendo and Parro (2015) to build a multiple-country and multiple-sector model with rich input–output linkages across different sectors.⁵ Then we rely on the most updated 2015 edition of the OECD Inter-Country Input–Output database (ICIO) to simulate the model. The OECD ICIO 2015 data provide a complete input–output matrix for the 34 ISIC Rev. 3 sectors of 61 countries and the rest of the world in 2011. Our sample covers 34 OECD countries and 17 non-OECD emerging economies.⁶ In particular, our sample includes BRICS (Brazil, Russia, India, China, and South Africa), the four Asian dragons (Korea, Taiwan, Hong Kong, and Singapore), the four Asian emerging tigers (Indonesia, Malaysia, the Philippines, and Thailand), and even low-income Asian countries like Cambodia and Vietnam. We drop the last sector (private households with employed persons) because this sector does not provide intermediate inputs for production in other sectors and its output is equal to 0 for half of the countries in our sample.⁷ In the end, we obtain a sample of N = 62 countries and J = 33 sectors (18 tradable sectors and 15 service sectors).⁸

The two data sets, OECD ICIO 2015 and OECD STAN ISIC REV3 2011, provide information on bilateral trade flows, bilateral expenditures, share of value-added, shares of

⁵ Please see Caliendo and Parro (2015) for details on basic setup and relative changes in equilibria of the model.

⁶ Please see OECD ICIO 2015 for the 61 countries and 34 sectors. The remaining sample countries that are not identified in the text. We also list the main 18 tradable sectors in the following tables for experiments on tariff changes.

⁷ Half of countries do not collect the data on this sector.

⁸ Athukorala and Khan (2016) suggest that the American relative prices of parts and components are remarkably less sensitive to changes in relative prices compared with that of final goods. For this reason, it would be beneficial to use even more disaggregated industrial data in future research.

materials (intermediate input) in production, and shares of final consumption for each country and each sector. The elasticity of substitution across sectors are taken from Table 1 in Caliendo and Parro (2015). Given these values, we can simulate the model and calculate the changes in output, trade flows, and welfare given tariff changes.

4. Quantifying the effects of tariff increases

4.1 Tariff increases

Because we use 2011 trade and production as our base year, the countries in our sample are all WTO members and impose MFN tariffs on one another. The sectoral mean or median of MFN tariffs are all less than 3 percent for all but three sectors—namely, agriculture (3.47 percent), food (8.07 percent), and textiles (8.77 percent). Therefore, we treat the initial tariff as equal to 0 for all countries and sectors.⁹

President Trump threatened to impose high tariffs of up to 45 percent on products imported from China in order to offset "currency manipulation". In this paper, we consider an extreme case in which the United States imposes this tariff on *all* imports from China. An alternative but equivalent interpretation is that the United States labels China as a currency manipulator and forces the Chinese currency to appreciate its currency by roughly 45 percent. Consider an increase from a zero tariff to a 45 percent U.S. tariff rate on all Chinese goods, $\hat{\tau}_{USA,CHN}^{j} = 1.45$. We follow Caliendo and Parro (2015) to solve for the equilibrium. First, we guess a vector of relative wages $\hat{\mathbf{w}}$, and then we plug wages in the equilibrium conditions to solve the changes in the unit cost for production $\hat{c}_n^j(\hat{\mathbf{w}})$ and good price $\hat{P}_n^j(\hat{\mathbf{w}})$. Second, we solve the changes in bilateral import shares $\pi_{ni}^{j'}(\hat{\mathbf{w}})$. Finally, we solve for the total expenditure in each sector $X_n^{j'}(\hat{\mathbf{w}})$, and then verify if the trade balance holds. If not, we adjust our guess $\hat{\mathbf{w}}$ until we achieve the equilibrium condition.

4.2 Sectoral bilateral trade between the United States and China

Before we discuss the effects of a tariff increase on trade flows and output, we discuss the relative tradability of U.S. and Chinese production across different sectors. Table 3 presents the China–U.S. bilateral trade flows in 18 tradable goods sectors in 2011. Particularly, the table presents the shares of bilateral import over the total imports and exports in each sector for the United States and China. The second column, $\frac{M_{USA,CHN}^{i}}{M_{USA}^{i}}$, provides the share of U.S. imports from China in sector *j* over the U.S. total imports in sector *j*. Two sectors, namely, computers and textiles, have the largest sectoral import shares—both above 45 percent and China is the United States' largest trading partner in each of these sectors. Electrical

⁹ Admittedly, China's current average import tariff is around 9 percent. Therefore, a hypothesized 45 percent high import tariff against China is similar to the effective 36 percent import tariff against the same country, which is a typical number of China's special safeguard tariffs imposed by the United States in past years.

Sector	^м ^j usa,сни	м ^j USA,CHN	^j CHN,USA	M ^j CHN,USA
	M_{USA}^{j}	E_{CHN}^{j}	M_{CHN}^{j}	E_{USA}^{j}
Agriculture	2.34	6.24	21.93	18.07
Mining	0.13	4.50	0.71	6.13
Food	7.63	15.17	13.61	7.69
Textiles	45.61	23.89	6.21	8.40
Wood	27.85	26.90	13.08	16.45
Paper	14.48	24.58	43.91	15.70
Petroleum	1.67	6.07	6.20	2.08
Chemicals	7.77	12.93	11.17	9.59
Plastics	25.88	25.82	6.77	6.64
Minerals	31.79	16.57	13.20	11.60
Basic metals	3.53	4.84	3.57	9.96
Metal prod.	28.23	19.92	11.01	5.25
Machinery n.e.c.	20.67	20.39	8.86	8.18
Computer	47.06	29.04	5.88	16.52
Electrical	31.18	21.61	6.02	11.61
Auto	5.43	23.47	8.17	5.73
Other transport	7.44	4.27	27.83	5.18
Others	30.02	24.83	15.55	2.76

Table 3. The sectoral bilateral trade flows in 2011 (percent)

Source: OECD ICIO (2015) and OECD STAN ISIC REV3 (2011).

Note: $\frac{M_{USA,CHN}^{j}}{M_{USA}^{j}}$ (or $\frac{M_{USA,CHN}^{j}}{E_{CHN}^{j}}$): imports of the United States from China in sector j over the total imports of the United States in sector j (the total exports of China in

sector j) in 2011.

equipment and minerals are the next two largest sectors of U.S. imports from China. Notably, these four sectors are also among the most important sectors of Chinese export to the United States. The third column, $\frac{M_{USA,CHN}^{j}}{E_{CHN}^{j}}$, shows the share of U.S. imports from China in sector *j* relative to Chinese total exports in the same sector. It indicates that China intensively exports to the United States in the computer, wood, plastic, papers, and textiles sectors. By contrast, China intensively imports from the United States in the paper, other transport (such as aircraft), and agriculture (fourth column) sectors. Moreover, about 18 percent of the total agricultural exports of the United States are consumed in China (fifth column). In sum, the strength of export capability in the United States and China varies across sectors.

For further perspective, Table 4 displays sector-level ratios of imports and exports relative to gross output and the relative output shares in the world for China and the United States. These data show that U.S. imports of textiles, computers, and electrical equipment are very high compared with domestic output. These goods are mainly exported by China (shown in Table 3). The U.S. imports from China as a share of all U.S. imports in these three sectors (the second column in Table 3) are all above the China's sectoral GDP share in world GDP (the last column in Table 4), respectively. The third column of Table 4 shows that U.S. export advantages are concentrated in in the other transport, machinery n.e.c., and computer sectors. On the production side the United States is responsible for more than 20 percent of the world output of the paper, petroleum, and other transport sectors.

	U.S.			China	China		
Sector	M_i^j/Y_i^j	E_i^j/Y_i^j	Y_i^j / Y_w^j	M_i^j/Y_i^j	E_i^j/Y_i^j	Y_i^j / Y_w^j	
Agriculture	7.51	14.48	8.02	3.86	0.91	25.28	
Mining	52.90	6.43	9.95	29.81	0.81	18.68	
Textiles	141.96	25.87	3.25	2.69	20.83	44.79	
Wood	15.49	7.26	8.37	1.79	3.14	42.66	
Paper	4.49	12.03	26.30	8.67	5.34	13.04	
Petroleum	11.80	15.53	20.56	7.24	4.52	14.85	
Chemicals	23.40	24.26	14.98	13.79	9.31	22.67	
Plastics	25.04	13.29	10.39	4.02	7.74	33.67	
Minerals	17.21	9.70	5.67	1.06	4.09	45.79	
Basic metals	33.99	12.72	7.23	6.77	4.73	37.82	
Metal prod.	13.79	10.78	14.39	3.74	14.23	19.77	
Machinery n.e.c.	43.87	36.64	9.11	9.65	12.67	31.97	
Computer	86.95	35.13	10.02	33.55	47.92	29.48	
Electrical	68.91	26.28	5.84	6.95	13.64	42.57	
Auto	42.42	21.10	12.00	7.93	5.25	22.40	
Other transport	14.38	37.82	20.08	8.04	28.60	17.60	

Table 4. Summary statistics on trade and output percent

Source: OECD ICIO (2015) and OECD STAN ISIC REV3 (2011).

Note: M_i^j/Y_i^j and E_i^j/Y_i^j denote the import share and export share in country i 's output, respectively, and Y_i^j/Y_w denotes the output share of country i in the world.

In contrast, China has a different trade structure and production pattern. First, China imports and exports heavily in a number of sectors including computers—which is suggestive of involvement in global value chains and processing trade. Second, China produces much more output than the United States in all sectors, aside from the paper, petroleum, and other transport sectors.

Based on Tables 3 and 4, we can draw three conclusions about China–U.S. production and trade patterns in 2011. First, the United States and China together produce more than 40 percent of the world tradable goods on average and specialize in different sectors. Second, the total trade of these two countries represents more than 20 percent of the world trade on average. Third, the trade in the textiles, computer, electrical equipment, machinery n.e.c., and other transport sectors is essential to understanding the China–U.S. trade relationship.

4.3 Case I: Unilateral U.S. tariffs with balanced trade

We first discuss how output and trade would be affected if the United States unilaterally imposes a 45 percent import tariff on Chinese goods, under the assumption that all countries achieve balanced trade afterward. In this case, China's exports to the United States face higher tariffs, but China does not raise its tariffs on its imports from the United States, which in fact facilitates trade rebalancing between the two countries. Table 5 presents our calibration results regarding changes in output, imports, exports, and bilateral trade between the United States and China.

First and as expected, U.S. imports from China in most sectors plummet. Bilateral imports in half of 18 sectors decline by more than 90 percent. This also leads to a significant decline

Sector	Y_{USA}^j	M_{USA}^{j}	E_{USA}^{j}	$M^{j}_{USA,CHN}$	Y^j_{CHN}	M^j_{CHN}	E_{CHN}^{j}	M ^j _{CHN,USA}
Agriculture	2.37	-8.04	7.29	-97.80	0.83	0.49	-1.63	8.57
Mining	12.31	-4.11	15.90	-99.55	2.22	-5.66	3.84	14.63
Food	-3.42	-11.03	1.94	-75.37	1.32	0.93	-10.12	3.31
Textiles	24.85	-29.34	4.84	-95.69	-6.51	-3.78	-21.30	1.24
Wood	5.46	-28.42	6.66	-99.06	-0.68	-3.90	-23.53	7.54
Paper	5.48	-19.57	14.01	-99.86	-2.84	1.10	-21.75	11.24
Petroleum	14.47	-45.05	60.96	-100.00	2.45	-26.62	17.27	61.40
Chemicals	1.85	-8.19	2.71	-78.54	-2.39	-2.93	-9.55	0.21
Plastics	4.94	-12.42	0.93	-61.17	-3.31	-3.28	-14.96	-1.94
Minerals	6.55	-18.63	2.09	-70.31	1.03	1.01	-10.56	2.99
Basic metals	6.81	3.07	2.40	-78.33	-0.87	-1.81	-2.41	0.25
Metal prod.	7.65	-24.63	5.08	-94.69	-3.09	-3.27	-16.94	3.49
Machinery n.e.c.	-3.05	-18.28	2.32	-62.37	-0.26	0.16	-11.30	1.18
Computer	31.84	-27.53	8.24	-96.05	-14.67	-7.68	-25.63	0.47
Electrical	22.24	-18.27	9.72	-99.32	-2.43	-4.82	-17.97	6.08
Auto	-0.28	-3.96	0.85	-65.33	0.55	0.38	-14.26	1.00
Other trans.	3.58	1.46	1.66	-37.59	1.03	1.51	-1.43	1.67
Others	-0.07	-27.89	3.00	-84.91	-4.83	0.07	-19.96	2.59
Average	7.98	-16.71	8.37	-83.11	-1.80	-3.23	-11.23	7.00

Table 5. Changes in trade and output: Case 1 (percent)

Note: Y_{USA}^{j} , M_{USA}^{l} , E_{USA}^{j} , $M_{USA,CHN}^{j}$ denotes the sector j's output, imports, exports, and imports from China in the United States.

in sectoral imports for the United States. For example, imports in the petroleum, textiles, wood, and computer sectors decline by more than a quarter. To compensate for the fall in imports, domestic production in the United States increases, particularly in the computer, textile, and electrical equipment sectors, which heavily rely on imports from China before the tariff hike.¹⁰ U.S. exports increase moderately through the effects of trade rebalancing.

By contrast, the high unilateral U.S. tariff hike has a catastrophic effect on Chinese exports, which fall by roughly 13 percent, with the greatest impacts occurring in the textile, wood, and computer sectors. This tariff shock also leads to a significant decline in China's sectoral gross output. For example, textile and computer sector output decreases by 6.51 percent and 14.67 percent, respectively. China's total imports also decrease heavily in the petroleum and computer sectors, by 26.62 percent and 7.68 percent, respectively. China's imports from the United States increase in 17 sectors because of trade rebalance. In particular, China's imports from the United States in the petroleum sector increase heavily because U.S. petroleum production increases 14.47 percent and also takes a very large share in the world petroleum (20.56 percent in Table 4).

¹⁰ Column Y^j_{USA} (L^j_{USA}) presents the changes in the U.S. output (labor). We use a Cobb-Douglas production function with labor and intermediate inputs for all sectors. The changes in sectoral labor inputs are equal to the output changes minus the changes in nominal wage. Because wage is equalized in all sectors within country, the changes in labor shares across different sectors in a country is proportional to the sectoral output changes. This result holds for all cases.

Rank	Name	w_n/P_n	Rank	Name	w_n/P_n
1	Singapore	2.58	53	France	-0.35
2	Luxembourg	2.17	54	Costa Rica	-0.37
3	Ireland	2.04	55	Cambodia	-0.39
4	Brunei	1.90	56	Romania	-0.51
5	Iceland	1.42	57	Tunisia	-0.57
6	Malaysia	1.40	58	India	-0.65
7	Switzerland	1.19	59	USA	-0.66
8	Norway	1.19	60	Portugal	-0.66
9	Saudi Árabia	1.12	61	Greece	-0.99
10	Netherlands	1.08	62	Turkey	-1.12
38	China	-0.04		5	

Table 6. Changes in real wages: Case 1 (percent)

Note: w_n/P_n denotes real wages in country n.

We use real wages to measure social welfare in each country. By this measure, Table 6 shows that the United States experiences a 0.66 percent welfare loss, and China also encounters a welfare loss—but of a much smaller magnitude, 0.04 percent. On first glance, this result may appear to be counterintuitive; because trade rebalance prevents U.S. residents from financing their consumption by borrowing abroad, however, the welfare decline occurs as U.S. consumption demand and real wages in turn are depressed. In contrast, the trade rebalancing benefits the Chinese as they do not need to save for other countries, which partially offsets the negative effect of trade war on Chinese domestic production. If we do not impose trade rebalance and assume that trade imbalance remains the same, then the United States real wage would decline by 0.28 percent and Chinese real wage would drop by 0.21 percent.

Although some countries are also harmed through their participation in global value chains and because of general equilibrium effects, some small countries, such as Singapore and Luxembourg, benefit as the unilateral tariff hike leads to trade diversion. China might increase its exports from those countries in response to the sharp decline in its exports to the United States. Finally, because the United States produces more and expands its exports, small importing countries will benefit from lower prices as the prices of goods in equilibrium fall.

4.4 Case 2: U.S.-China retaliatory tariff war with balanced trade

Next, we consider the case when China chooses to retaliate by increasing tariffs to 45 percent on its imports from the United States. As with the first scenario, we assume that all countries achieve trade balance afterward. Our calibration results in Table 7 show that bilateral trade between the United States and China collapses because of the trade war. Bilateral imports in half of the 18 sectors drop by more than 90 percent. The collapse of trade is particularly pronounced in the sectors where countries have their comparative advantage—U.S. exports of agriculture, wood, paper, and computers, and China's exports

Sector	Y_{USA}^j	M_{USA}^{j}	E_{USA}^{j}	$M^{j}_{USA,CHN}$	Y^j_{CHN}	M_{CHN}^j	E_{CHN}^{j}	$M^{j}_{CHN,USA}$
Agriculture	-1.14	-10.67	-10.12	-97.94	2.45	-18.69	-4.84	-97.27
Mining	14.05	-4.75	11.64	-99.57	1.93	-2.75	-0.27	-99.44
Food	-4.18	-11.85	-3.47	-75.81	2.28	-7.84	-10.80	-72.45
Textiles	23.80	-30.31	-1.40	-95.84	-6.29	-7.72	-22.47	-96.40
Wood	3.75	-30.15	-9.12	-99.11	0.38	-14.44	-25.56	-98.90
Paper	3.12	-22.26	-0.26	-99.88	2.30	-41.50	-25.71	-99.81
Petroleum	16.51	-50.34	72.33	-100.00	2.32	-26.74	2.23	-100.00
Chemicals	-0.30	-9.58	-4.20	-79.08	-0.67	-9.16	-10.28	-77.61
Plastics	4.02	-13.27	-2.94	-61.73	-2.46	-6.12	-15.42	-62.96
Minerals	5.43	-19.47	-6.04	-70.80	1.69	-7.64	-11.04	-70.45
Basic metals	4.72	1.35	-5.20	-78.88	-0.13	-3.21	-2.98	-79.13
Metal prod.	6.48	-26.16	1.03	-94.89	-2.35	-11.83	-18.20	-94.46
Machinery n.e.c.	-4.52	-18.98	-2.32	-62.84	0.56	-3.90	-11.66	-58.59
Computer	27.49	-29.13	-4.97	-96.24	-14.26	-9.67	-26.98	-96.88
Electrical	19.87	-19.95	0.11	-99.36	-1.95	-8.06	-19.90	-99.35
Auto	-1.27	-4.65	-2.89	-65.76	1.42	-3.66	-14.72	-64.25
Other trans.	3.05	0.89	-0.34	-38.04	1.60	-8.99	-1.55	-38.69
Others	-0.60	-28.69	1.50	-85.29	-4.13	-11.03	-21.01	-83.27
Average	6.68	-18.22	1.85	-83.39	-0.85	-11.27	-13.40	-82.77

Table 7. Changes in trade and output: Case 2 (percent)

Note: See notes to Table 5.

Table 8. Changes in real wages: Case 2 (percent)

Rank	Name	w_n/P_n , percent	Rank	Name	w_n/P_n , percent
1	Singapore	2.63	53	France	-0.35
2	Luxembourg	2.17	54	Costa Rica	-0.37
3	Ireland	2.04	55	Cambodia	-0.40
4	Brunei	1.93	56	Romania	-0.51
5	Malaysia	1.47	57	Tunisia	-0.57
6	Iceland	1.42	58	India	-0.65
7	Switzerland	1.19	59	Portugal	-0.67
8	Norway	1.17	60	USA	-0.75
9	Saudi Árabia	1.13	61	Greece	-1.00
10	Netherlands	1.07	62	Turkey	-1.12
37	China	0.08		5	

Source: Authors' calculations.

Note: w_n/P_n denotes real wages in country n.

of textile, computer, and electrical products. Overall, the imports of the United States and China decrease by 17 percent and 6 percent, respectively. Although the declines in imports in the two countries are consistent, their exports show sharp differences because of trade rebalancing. U.S. overall exports rise by 9 percent to achieve trade balance, whereas Chinese exports fall by 15 percent.

In this scenario, the United States faces more severe challenges than China because it needs to boost its exports to restore trade balance. As shown in Table 8, the real wage in the United States decreases by 0.75 percent, which represents a larger welfare loss than the first case. By contrast, China actually gains slightly. In this case the real wage increases by 0.08 percent, because the income effect driven by trade rebalancing outweighs the negative effect of an import tariff hike in the United States.

Sector	Y_{USA}^j	M_{USA}^j	E_{USA}^{j}	$M_{USA,CHN}^{j}$	Y^j_{CHN}	M^j_{CHN}	E_{CHN}^{j}	$M^{j}_{CHN,USA}$
Agriculture	-3.97	-2.13	-20.56	-97.49	0.24	-28.55	0.66	-97.84
Mining	-3.49	-0.94	-9.41	-99.50	2.46	-6.56	3.63	-99.59
Food	-0.27	-4.78	-6.58	-73.30	-0.56	-12.17	-9.07	-74.62
Textiles	20.83	-19.80	-12.03	-95.05	-5.65	-13.20	-17.45	-96.97
Wood	1.16	-20.36	-20.66	-98.90	-1.12	-21.54	-18.13	-99.14
Paper	-3.14	-9.81	-20.36	-99.83	4.75	-51.70	-14.86	-99.86
Petroleum	-3.02	5.11	-9.93	-100.00	4.22	-40.04	35.44	-100.00
Chemicals	-1.20	-4.31	-8.55	-77.70	-1.45	-11.27	-8.46	-79.15
Plastics	1.84	-13.25	-4.32	-61.40	-3.39	-7.59	-15.26	-64.46
Minerals	1.99	-19.40	-9.42	-70.31	-1.46	-12.05	-10.56	-72.78
Basic metals	0.00	-0.45	-8.81	-78.72	-1.68	-6.11	-2.30	-80.58
Metal prod.	2.22	-21.45	-6.82	-94.33	-3.07	-16.18	-14.55	-95.26
Machinery n.e.c.	-0.07	-8.97	-5.70	-57.96	-1.48	-6.81	-10.12	-60.77
Computer	12.19	-25.21	-19.23	-95.98	-13.14	-11.53	-24.86	-97.39
Electrical	4.59	-15.29	-16.75	-99.27	-2.71	-13.20	-15.40	-99.50
Auto	-0.65	-1.81	-4.71	-64.24	-1.06	-6.72	-14.05	-66.23
Other trans.	-0.43	-2.78	-2.50	-40.11	-0.30	-11.63	-1.76	-40.70
Others	3.58	-17.61	-3.63	-82.40	-4.06	-16.40	-16.30	-85.02
Average	1.79	-10.18	-10.55	-82.58	-1.64	-16.29	-8.52	-83.88

Table 9. Changes in trade and output: Case 3 (percent)

Note: See table notes in Table 5.

Table 10. Changes in real wages: Case 3 (percent)

Rank	Name	w_n/P_n	Rank	Name	w_n/P_n
1	Cambodia	0.22	53	Canada	-0.01
2	Costa Rica	0.11	54	South Africa	-0.01
3	Singapore	0.09	55	Korea	-0.02
4	Vietnam	0.08	56	Saudi Arabia	-0.02
5	Mexico	0.06	57	Australia	-0.02
6	Israel	0.05	58	Brunei	-0.03
7	Cyprus	0.04	59	Chile	-0.03
8	Italy	0.04	60	Hong Kong	-0.04
9	Taipei	0.03	61	USA	-0.32
10	Estonia	0.03	62	China	-0.37

Source: Authors' calculations.

Note: w_n/P_n denotes real wages in country n.

4.5 Case 3: U.S.-China retaliatory tariff war with ongoing trade imbalance

The previous two cases show that trade rebalancing plays an important role in reshaping the trade pattern, output, and real wage for the United States and China. Thus, we consider the third case where trade remains unbalanced and the United States and China both increase their bilateral import tariffs against each other to 45 percent. More specifically, we assume the United States maintains trade deficits and China remains in trade surplus as before the trade war. Tables 9 and 10 present our calibration results, demonstrating three differences compared with the case 2 in previous subsection.

First, although bilateral imports decrease dramatically as in case 2, in case 3 China suffers more than the United States, as is shown in Table 11. The reason is that China needs to maintain the trade surplus as before and the United States can still increase its consumption and imports by running trade deficits. Second, U.S. exports increase in case 2 to restore

	China			United States			
Var.	Case1	Case2	Case3	Case1	Case2	Case3	
Output	-0.68	-0.07	-1.36	-0.72	-1.08	-0.11	
price index	0.01	0.38	-0.42	-0.74	-0.98	0.42	
Exports	-12.96	-14.70	-10.61	11.39	9.25	-6.68	
Imports	-4.17	-6.09	-10.60	-15.05	-16.68	-6.70	
Nominal wage	-0.03	0.46	-0.79	-1.39	-1.73	0.10	
Real wage	-0.04	0.08	-0.37	-0.66	-0.75	-0.32	

Table 11. Comparison of the three tariff policies (percent)

Note: This table compares the output, price index, and trade for China and the United States for the three policy experiments.

trade balance despite the trade war, whereas in case 3 they are depressed by the tariff hike in China.

The United States and China both lose in this case and they become the two largest losers in the trade war. The real wage in China and the United States decreases by 0.37 percent and 0.32 percent, respectively. China experiences larger losses in this case because the country needs to maintain its large trade surplus while losing its largest market (i.e., the United States). By contrast, the United States slightly improves compared with its situation in case 2 because the United States can still maintain its consumption through external borrowing by running trade deficits. If, however, we take into account China's future return on current savings and the United States' future payment on its current debt, the welfare loss would be smaller for China and larger for the United States.

5. Concluding remarks

This paper examines the possible effects of a U.S.–China trade war on international trade, output, and social welfare. Through the use of a standard multi-country and multi-sector general equilibrium model we evaluate the effects of high U.S. tariffs, as were proposed in Trump's campaign for the U.S. presidency. We simulate three different scenarios depending on whether China chooses to retaliate and whether the U.S.–China trade imbalance is eliminated. In any of the scenarios we find that the trade war will lead to a collapse in bilateral U.S.–China trade. As a consequence, the United States will suffer from social welfare losses; China will also experience social welfare losses if the trade imbalance remains. The first two case, however, imply that China's best defensive move—given the risk of a trade war with the United States—is to reduce the total trade surplus and the share of imports it purchases from the United States.

Thus, our findings have important policy implications. On one hand, the United States should try to credibly commit to continued WTO-based trade with China because the tariff threat against China leads to further deterioration of the real wage and welfare. On

the other hand, China's focus on "One-Belt-One-Road" (OBOR) countries might be particularly fruitful given China's already-large trade volumes with these countries. Consequently, China can increase its import from non-U.S. countries to reduce the potential losses tied to a trade war with the United States.

Two possible extensions merit special consideration. The first is the evaluation of trade policy in the context of regional trade agreements and regional integration. In particular, China is currently active in the pursuit of regional trade agreements, such as the ongoing Regional Comprehensive Economic Partnership and the OBOR initiative. For this reason, if the United States were to launch a protectionist action against China, it might also take action against China's associated trading bloc partners. In turn, the trade war could expand further if China and its partners were to impose trade barriers not only on the United States, but also in the trade bloc partners of the United States. Second, this paper does not discuss the exchange rate adjustments in which would be triggered by a trade war, but which may play an important role in reshaping the trade imbalance (Woo 2008).¹¹ Nonetheless, our case studies under a number of trade balance scenarios may still shed light on the consequences of a trade war which coincides with an ongoing trade imbalance or a subsequent trade balance. A careful treatment of this question will require the endogenization of the exchange rate and trade imbalance. The analysis of this question is reserved for future research.

References

Athukorala, Prema-chandra, and Fahad Khan. 2016. Global Production Sharing and the Measurement of Price Elasticity in International Trade. *Economics Letters* 139:27–30.

Caliendo, Lorenzo, and Fernando Parro. 2015. Estimates of the Trade and Welfare Effects of NAFTA. *The Review of Economic Studies* 82(1):1–44.

Dekle, R., J. Eaton, and S. Kortum. 2008. Global Rebalancing with Gravity: Measuring the Burden of Adjustment. *IMF Economic Review* 55(3):511–540.

Di Giovanni, Julian, Andrei A. Levchenko, and Jing Zhang. 2014. The Global Welfare Impact of China: Trade Integration and Technological Change. *American Economic Journal: Macroeconomics* 6(3):153–183.

Donaldson, David. 2010. Railroads of the Raj: Estimating the Impact of Transportation Infrastructure. National Bureau of Economic Research Working Paper.

Eaton, Jonathan, and Samuel Kortum. 2002. Technology, Geography, and Trade. *Econometrica* 70(5):1741–1779.

Galle, Simon, Andres Rodriguez-Clare, and Moises Yi. 2015. Slicing the Pie: Quantifying the Aggregate and Distributional Effects of Trade. Working Paper. University of California, Berkeley.

¹¹ We thank Professors Wing Thye Woo and Fukunari Kimura for these insightful suggestions.

Petri, Peter A., Michael G. Plummer, and Fan Zhai. 2012. *The Trans-Pacific Partnership and Asia-Pacific Integration: A Quantitative Assessment*. Policy Analyses in International Economics Vol. 98. Washington, DC: Peterson Institute.

Tombe, Trevor, and Xiaodong Zhu. 2015. Trade, Migration and Productivity: A Quantitative Analysis of China. University of Toronto Working Paper No. 542.

Woo, Wing Thye. 2008. Understanding the Sources of Friction in US–China Trade Relations: The Exchange Rate Debate Diverts Attention from Optimum Adjustment. *Asian Economic Papers* 7(3):61–95.

VoxChina A Bridge on Economic Issues between China and the World

Evaluating the Burden of a U.S.-China Trade War

Meixin Guo, Lin Lu, Liugang Sheng, Miaojie Yu Apr 25, 2018

Trade disputes between the United States and China greatly intensified recently as the two countries announced a 25 percent tariff hike on \$50 billion worth of products imported from each other, raising the risk of a trade war between the two giant trading economies. Based on a standard multi-sector, multi-country general equilibrium trade model with input-output linkages, we evaluate the cost of a trade war in which the United States and China both increase their tariffs to 45% for all imports from each other. We find that the United States would be more likely to be the bigger loser and that the cost for China would be moderate.

The world trade system has been facing grave danger in 2018 ever since U.S. president Donald Trump threatened a 25 percent import tariff on \$50 billion worth of products from China, and China threatened to retaliate with a tit-for-tat tariff on imports from the United States. These two countries, which are the economic engines for the world economy and the largest exporters and importers in the world, are heading towards a high-stakes trade war. The risk of a U.S.-China trade war has significantly depressed global financial markets as increasing volatility has shadowed the economic outlook.

In our recent work (Guo et.al, 2018), we evaluate the possible impacts of a U.S.-China trade war on trade, output, and real wages by using a multi-sector, multi-country general equilibrium model with intersectional linkages. The model is an extension of Eaton and Kortum (2002), and the specific setting follows Caliendo and Parro (2015). The quantitative trade model provides a parsimonious approach to evaluating the quantitative consequences of changes in trade policy, and is an alternative to the traditional Computational General Equilibrium (CGE) model, which requires fully specified preferences, technology, and trade costs with many ad hoc parameters.

Our model covers 34 sectors in 62 economies including 34 OECD countries, 17 non-OECD emerging economies, and the rest of the world. We consider a hypothetical trade war in which the United States imposes a 45 percent import tariff for all imported goods from China. We choose this number because Trump proposed imposing 45 percent import tariffs during his meeting with the editorial board of The New York Times in January 2016.

VoxChina A Bridge on Economic Issues between China and the World

One important limitation of quantitative trade models is that a country's aggregate trade balance (not bilateral trade balance) is usually exogenously determined by assumptions (Ossa, 2016). Countries with trade deficits receive a net income transfer from countries with trade surplus. For simplicity, we consider two possibilities: the trade war restores trade balance or maintains the current trade imbalance. The reality would be in between, but these two possible scenarios explain why the United States is at a disadvantage: it will lose the net income transfer from its current trade deficits if trade rebalances after a trade war.

We consider three possible scenarios of a U.S.-China tariff war in the model simulation. In the first scenario, the United States increases its import tariffs to 45 percent on all imports from China and all countries achieve balanced trade after the trade war. Balanced trade might well be one of the goals of a trade war as the U.S. government has for a long time blamed China for its large trade surplus. In the second case, we assume China retaliates by increasing its tariffs to 45 percent on its imports from the United States as well, again evaluating the impacts under the assumption of balanced trade. In the third case, we consider a situation in which both China and the United States impose 45 percent tariffs on each other, but trade imbalances remain unchanged for all countries. For simplicity, we name these three cases: 1) Unilateral U.S. tariffs with balanced trade, 2) U.S.-China reciprocal tariff war with balanced trade, and 3) U.S.-China reciprocal tariff war with pre-existing trade imbalance, respectively.

Our exercise shows that in all scenarios, high import tariffs lead to a catastrophic collapse in bilateral trade between the U.S and China. In the first case, China's exports to the United States plummet upon the unilateral U.S. tariff hike, while the effects on U.S exports to China is rather moderate. In the second and third cases of a tit-for-tat tariff war, the bilateral imports between the two countries slump in all sectors, with half of the 18 sectors considered dropping by more than 90 percent. The collapse of bilateral trade is particularly pronounced in the sectors in which each country has a comparative advantage: U.S. exports of agricultural goods, wood, paper, and computers, and China's exports of textiles and computer and electrical products.

A U.S.-China trade war would generate substantial losses in output and social welfare measured as real wages. The United States is likely to be the biggest loser in all scenarios. In the worst case, its output and real wages would drop by 1.08 percent and 0.75 percent respectively (case 2 in **Table 1**). The losses for China in the first two cases with balanced trade would be very small. The comparison between the first two cases also indicates that

China would be better off if it takes retaliation when the U.S. increases its tariffs against China. In the third case with unbalanced trade, China and the United States would experience similar losses in social welfare, i.e., the real wage in the two countries would decrease by 0.37 percent and 0.32 percent respectively (case 3 in **Table 1**). In all cases, some Asian countries may gain slightly from the trade diversion, while many advanced economies would experience collateral damage due to spillover effects through input-output linkages and general equilibrium effects.

	China			United States			
Var.	Case1	Case2	Case3	Case1	Case2	Case3	
Output	-0.68	-0.07	-1.36	-0.72	-1.08	-0.11	
price index	0.01	0.38	-0.42	-0.74	-0.98	0.42	
Exports	-12.96	-14.70	-10.61	11.39	9.25	-6.68	
Imports	-4.17	-6.09	-10.60	-15.05	-16.68	-6.70	
Nominal wage	-0.03	0.46	-0.79	-1.39	-1.73	0.10	
Real wage	-0.04	0.08	-0.37	-0.66	-0.75	-0.32	

Table 1: Three Cases of U.S.-China Trade War

Source: Authors' calculations.

Note: This table compares the output, price index, and trade for China and the United States for the three policy experiments.

The costs of a trade war are asymmetric for two reasons. First, the quantitative model itself is asymmetric in many settings including the technology level, bilateral trade costs, and labor endowments. Moreover, the sectoral composition of each country is different and the sectoral responses to tariff hikes vary due to the different substitution elasticities. In the third case where the trade imbalance remains, the real wage declines for both countries but the mechanisms are very different. China loses the U.S. market due to high tariffs, and thus the exports and outputs drop more, which leads to a decline in nominal wages. However, prices in China also decline as the country produces more for itself. The real wage drops because the nominal wage declines more than the price adjustment. In contrast, the United States imports less and thus the price level increases. The nominal wage slightly increases as the country needs to hire more workers to meet domestic demand. Overall, the real wage in the United States also declines, but the mechanism is quite different from that of China.

Second, the assumption about trade balance also plays a crucial role in the asymmetric costs of the potential U.S.-China trade war. Currently the United

VoxChina A Bridge on Economic Issues between China and the World

States has large trade deficits, indicating higher expenditures than income. Restoring trade balance implies that the country should reduce its expenditures and therefore consumption. This explains why, in the first two cases, the U.S. would be at a disadvantage, and China may gain slightly from the trade war if trade rebalances.

In the first two cases in which the trade war would lead to a trade balance, the United States must export more and import less to move from a trade deficit to a trade balance, while China would need to reduce its exports more than it would need to reduce imports (cases 1 and 2 in Table 1). In other words, the United States would have to reduce its consumption expenditures, which acts a negative income shock. In contrast, the reduction in China's trade surplus acts as a positive income transfer to China, and thus would alleviate the negative effect of a trade war on its economy. Thus, in these two cases, social welfare loss would be even larger for the United States than China. However, if the trade war did not change the current trade imbalance, the United States would still spend more than its income, while China would consume less than its income. Thus, China may be hurt more than the United States by the trade war (case 3 in Table 1). However, this case is less likely to happen as a trade war would reduce the volume of trade, and thus the trade imbalance between two countries would be more likely to shrink. Moreover, our analysis implies that trade balance might not be a desirable target for the United States if it eventually triggers a trade war against China.

No one will win if two countries wage a trade war. Our calculation of welfare loss is rather conservative and likely to underestimate the effects of a possible trade war on outputs and social welfare. One key assumption in our model is that all economies function well without any other frictions, aside from trade costs, such as perfect labor mobility across sectors and financial frictionless. However, in reality, these adjustments may not be smooth, and the impact of a trade war on the world economy would be magnified. Moreover, a trade war is likely to trigger a tsunami in global financial markets, leading to a significant wealth loss.

The policy implication of our article is straightforward. China must consider how to deal with the enhanced trade disputes trigged by Trump. The best policy response would be to retaliate on the same scale against U.S. protectionist policies. More importantly, to guarantee that China is able to minimize its welfare loss, it is essential for China to simultaneously open up its domestic markets for China's most important trading partners like the EU and the ASEAN 10 countries. Specifically, China can import more from the EU and the one-belt-one-road countries to reduce its domestic prices and hence increase real income.

(Meixin Guo, School of Economics and Management, Tsinghua University; Lin Lu, School of Economics and Management, Tsinghua University; Liugang Sheng, Department of Economics, the Chinese University of Hong Kong; Miaojie Yu, National School of Development, Peking University.)

References

Caliendo, Lorenzo and Parro, Fernando. 2015. Estimates of the Trade and Welfare Effects of NAFTA. *The Review of Economic Studies*, Volume 82(1): Pages 1-44.

https://academic.oup.com/restud/article/82/1/1/1547758

Eaton, Jonathan and Kortum, Samuel. 2002. Technology, Geography, and Trade.*Econometrica*, Volume 70(5): Pages 1741-1779. https://onlinelibrary.wiley.com/doi/abs/10.1111/1468-0262.00352

Guo, Meixin, Lin Lu, Liugang Sheng, and Miaojie Yu. 2018. The Day After Tomorrow: Evaluating the Burden of Trump's Trade War. *Asian Economic Papers* Volume 17(1): 101-120.

https://www.mitpressjournals.org/doi/abs/10.1162/asep_a_00592

Ossa, Ralph. 2016. Quantitative models of commercial policy. *Handbook of Commercial Policy*, edited by Bagwell, K. and R. Staiger, Volume 1: 207-259. North-Holland.

https://www.sciencedirect.com/science/article/pii/S2214312216300035

Tags: <u>Trade Disputes</u>

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

China's evolving role in global production networks: Implications for Trump's trade war

P Athukorala - China's New Sources of Economic Growth: Human ..., 2017 - library.oapen.org The early literature on China's rise as an export powerhouse widely interpreted the shift in its export composition away from standard labour-intensive products towards 'high-tech'product lines within global production networks as an indication of China becoming an advanced ... $\frac{1}{29}$ Dited by 26 Related articles All 13 versions

Understanding the China–US trade war: causes, economic impact, and the worst-case scenario

TTL Chong, X Li - Economic and Political Studies, 2019 - Taylor & Francis This paper studies the current trade war between China and the US from a historical standpoint. By comparing the ongoing trade war with similar trade conflicts in history, we reveal three major causes, with varying degrees of importance, from both economic and ... $\frac{1}{29}$ Cited by 23 Related articles All 8 versions

Trade protectionism and US manufacturing employment

C Li, J Whalley - Economic Modelling, 2020 - Elsevier This paper builds a numerical global general equilibrium model to simulate the effects of USinitiated trade protection measures on manufacturing employment. The simulation results show that US trade protection measures reduce manufacturing employment in the US, and ... $\Delta \qquad 99$ Cited by 9 Related articles All 9 versions

The US–China trade war: Tariff data and general equilibrium analysis

M Li, EJ Balistreri, W Zhang - Journal of Asian Economics, 2020 - Elsevier The current trade war between the United States and China is unprecedented in modern history. This study introduces a database of tariff increases resulting from the trade war and quantifies the impacts using the canonical GTAPinGAMS model calibrated to the recently ... $\frac{1}{29}$ Dited by 7 Related articles All 9 versions

China's Free Trade Ports: Effective Action Against the Threat of De-globalization

W Tian, Z Xu, M Yu, H Zhu - China & World Economy, 2018 - Wiley Online Library With sluggish external demand and increasing trade protectionism by the USA and the European Union, China is facing severe challenges in implementing its deeper, ongoing reforms. To respond actively to such challenges, the Communist Party of China's 19th ... 39 Cited by 6 Related articles All 6 versions

Understanding the recent Sino-US trade conflict

M Yu, R Zhang - China Economic Journal, 2019 - Taylor & Francis To understand Sino-US trade relations, this article interprets the trade imbalance between China and the United States from the Trump administration's perspective. The Trump administration claims that the Chinese government's subsidies and high import tariffs cause ... $\Delta \qquad \overline{22}$ Cited by 6 Related articles All 4 versions

An analysis of the China–US trade war through the lens of the trade literature

LD Qiu, C Zhan, X Wei - Economic and Political Studies, 2019 - Taylor & Francis The current China–US trade war that started in early 2018 has been the largest of this kind in the global market in the past half century, if not longer. Many speculations about the reasons for and progress and potential implications of the trade war emerge. Countries must ... $\Delta \qquad \overline{22}$ Cited by 5 Related articles All 8 versions

Why will Trump lose the trade war?

L Sheng, H Zhao, J Zhao - China Economic Journal, 2019 - Taylor & Francis The escalating US–China trade conflicts have increasingly shadowed the outlook of the world economy. The Trump administration aims to achieve its strategic goals including reducing current account deficits, promoting the US manufacturing sector, and curbing ...

 $\cancel{2}$ $\cancel{2}$ Cited by 9 Related articles

Measuring the Benefits of the "One Belt, One Road" Initiative for Manufacturing Industries in China

J Hou, SC Chen, D Xiao - Sustainability, 2018 - mdpi.com

China has shifted its focus from export-led growth to technology-driven growth. The "One Belt, One Road" initiative (OBOR) is aimed at promoting economic integration with the rest of the world that might boost China's ability to upgrade its manufacturers. We evaluate the ... $\sum_{n=1}^{\infty}$ $\overline{99}$ Cited by 3 Related articles All 8 versions \gg

US-China Trade War: Potential Trade and Investment Spillovers into Malaysia

S Yean Tham, AKJ Yi, TB Ann - Asian Economic Papers, 2019 - MIT Press

This study examines the potential impact of the current trade war between the United States and China on Malaysia's trade and investment. For solar exports, the immediate impact from 2017 to 2018 shows that gross and domestic exports to the United States have fallen but re ... $\frac{1}{2}$ $\frac{5}{2}$ Cited by 5 Related articles All 8 versions

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

中美贸易摩擦的缘起与对策——一个文献综述 余淼杰,金洋,刘亚琳-长安大学学报(社会科学版),2018 - nsd.pku.edu.cn v" 7# 4Ô£¤ c øý, \$% 2018Ê8 &êü \$» ïc<= Ów, Cdtu# 59';()* c+ Y, 7# 4Ô1 · 5,\Ü GH8Ú7# · 5T c-..<= vw÷/, 7# 4Ô1# 5,\Ü GHN0M¶ S c,# 5¼1V Vc2A¦ 345 7# 4Ô. ½PĨ¢, 6Y4Ôc7c89;(T,# 5TU § " cT N 5Øc: Mt;» ¼3Z,< Ò § " cЇ: A=.-rÃc4Ô89;(X> 1# 53Y1Ó § -c?;@ A. 75TUú"-. ĭ1 Bc» ½... ☆ 99 Cited by 2 Related articles All 7 versions ≫

Can the Regional Comprehensive Economic Partnership minimise the harm from the United States–China trade war?

R Mahadevan, A Nugroho - The World Economy, 2019 - Wiley Online Library Using ex post tariff schedules for the first time, it was found that the global gains provided by the Regional Comprehensive Economic Partnership (RCEP) are not enough to overcome the negative impacts of the United States–China trade dispute. While trade tensions cause ... $\frac{1}{29}$ <u>Sited by 2</u> Related articles All 7 versions

Evaluating the Impact of the US–China Trade War on Euro Area Economies: A Tale of Global Value Chains

I Fusacchia - Italian Economic Journal, 2019 - Springer

The US–Sino trade war will have significant repercussions on the global trade system. This study assesses the implications of the successive rounds of tariff hikes implemented by the US and Chinese governments on global value chains, with a focus on the major European ... $\frac{1}{22}$ $\frac{52}{22}$ Cited by 1 Related articles All 2 versions

CHINA'S ROLE IN INTERNATIONAL TRADE AMID DURING IN TRADING ATTACKS WITH THE UNITED STATES

C Dewi - PEOPLE: International Journal of Social Sciences, 2019 - grdspublishing.org Protectionist policies towards Chinese trade aim to balance US trade with China. The United States has seen the Chinese economy which has expanded into the world and become a new emerging force, even the United States sees the Made in China plan in 2025. Amid ... \therefore Σ Cited by 1 Related articles \Longrightarrow

The Status of China's Market Economy and Structural Reforms: The Issues Behind the

US–China Trade War

M Yu - Asian Economic Papers, 2019 - MIT Press

This paper investigates the most essential issues behind the ongoing US–China trade war. In addition to the apparent bilateral trade imbalance, China's status as a non-market economy (as labelled by the Trump administration) is one of the most fundamental reasons ... $\stackrel{\wedge}{\longrightarrow}$ $\stackrel{\nabla}{\cancel{D}}$ Cited by 3 Related articles All 5 versions

Exploring the impacts of Sino–US trade disruptions with a multi-regional CGE model

L Cui, Y Sun, R Melnikiene, M Song... - Economic research ..., 2019 - hrcak.srce.hr Sažetak With the aim to explore the boundary effects of the Sino–US trade war, this study considers a multi-region CGE model to set up six trade disruption scenarios based on the severity of trade friction, and empirically examines the gains and losses for China and the ... Σ_{T} Σ_{T} Cited by 3 Related articles All 6 versions

International Journal of Law, Humanities & Social Science©

E Mgeel-Kafui - academia.edu

The ongoing trade war between the United States and China with the view of offering a legal justification for the additional tariff measures adopted by China under the WTO legal system. The study utilizes a critical literature review study design in establishing the legal options ...

 $\cancel{2}$ $\cancel{9}$ Related articles All 3 versions $\cancel{8}$

Nuclear Fusion Diffusion: Theory, Policy, Practice, and Politics Perspectives

EG Carayannis, J Draper... - IEEE Transactions on ..., 2020 - ieeexplore.ieee.org Fusion energy has the potential to address long-term energy requirements and climate change. However, fusion energy is characterized by a "development divide" between the International Thermonuclear Experimental Reactor (ITER) consortium members and the non ...

 $\cancel{2}$ $\cancel{99}$ Cited by 1 Related articles

Multinational Companies' Hedging Effectiveness of Foreign Exchange Risk: A Quantitative Comparison Study

X Zhang - Fudan Journal of the Humanities and Social Sciences, 2020 - Springer

To investigate hedging effectiveness of multinational companies in respect of using currency derivatives, the author adapts an innovative and multi layers GJR-GARCH-based model. This model broke down the currency risk faced by MNCs in each business area and added ... $\frac{1}{2}$

Global impacts of China–US trade tensions

PS Tam - The Journal of International Trade & Economic ..., 2020 - Taylor & Francis This paper studies the economic impacts of China–US trade tensions. A global value chainsaugmented global vector autoregressive trade model that gauges the spatial and temporal dynamics of international transmission of economy-specific trade policy shock across ...

 $\cancel{2}$ $\cancel{9}$ Related articles

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

MODERN TRADE WARFARE: TRADE WAR RESPONSE AND CONTINGENCY FRAMEWORKS

HT Ganter, AL Kjøller, JH Mortensen - 2020 - research-api.cbs.dk

A recent rise in protectionism has led to an increase in trade disputes and trade wars which cause changes to the trade environment. These changes often cause negative impacts on multinational companies (MNCs), which MNCs are strategically ill-prepared to cope with ... $\frac{1}{22}$ $\frac{5}{22}$ All 2 versions $\frac{1}{22}$

US-China Trade War: Is Winter Coming for Global Trade?

X Tu, Y Du, Y Lu, C Lou - Journal of Chinese Political Science, 2020 - Springer

This paper uses the SMART Model to analyze the direct economic effects of the US-China tariff war on both China and the US. Based on the three lists of Chinese products subject to additional tariffs imposed under the US Section 301 investigation throughout 2018 and the ...

 $\cancel{2}$ $\cancel{99}$ Cited by 2 Related articles

China's Free Trade Ports: An Effective Way Against the Threat of De-Globalization

W Tian, Z Xu, M Yu, H Zhu - ccer.pku.edu.cn With sluggish external demand and heating trade protectionism in the United States and the European Union, China's deeper, ongoing reforms face severe challenges. To respond actively to such challenges, the Communist Party of China's 19th National Congress ...

 $\cancel{2}$ $\cancel{99}$ Related articles All 8 versions $\cancel{88}$

The effects of the China–US trade war during 2018–2019 on the Chinese economy: an initial assessment

K Liu - Economic and Political Studies, 2020 - Taylor & Francis

The China–US trade war during 2018–2019 has attracted attentions from academics, policy makers, businesses and investors around the world. Unlike previous researches which are mainly based on hypothetical scenarios, this study looks at the real effects of the China–US ... $\sum_{n=1}^{\infty}$ $\sum_{n=1}^{\infty}$ Cited by 1 Related articles

X 22 Clied by I Related articles

Processing Trade, Trade Liberalization, and Opening-Up: China's Miracle of International Trade

M Yu - China-US Trade War and Trade Talk, 2020 - Springer

Understanding the role of international trade is the key to understanding China's miraculous economic growth. This paper reviews the literature on international trade in the context of China, with a focus on processing trade, trade Liberalization, and firm performance to ... $\frac{1}{\sqrt{2}}$ $\frac{50}{2}$ Related articles All 4 versions

International Balkan and Near Eastern Social Sciences Congress Series XI. IBANESS Congress Series-Tekirdağ/TURKEY

DK DIMITROV, D NIKOLOSKI, R YILMAZ - 2019 - researchgate.net The priorities as set in Europe 2020 of fostering development of smart, sustainable and inclusive economy are interconnected and all related to the field of education, research and innovation woven in the goals of employment, innovation, education, poverty reduction and ...

 $\cancel{2}$ $\cancel{99}$ Related articles All 4 versions $\cancel{88}$

Imposing tariff or implementing subsidy?

Z Chen, SII Su, H Wang - Journal of Modelling in Management, 2020 - emerald.com Purpose In the context of the trade war in full swing, the global supply chain systems have experienced a serious shock and become very vulnerable. The purpose of this paper is to explore the intertwining effects between the export-supporting subsidy policy and the import ...

 $\cancel{2}$ $\cancel{99}$ Related articles

Understanding Sino–US Trade War: An American Government Perspective

RS Bhandari, S Bansal, LK Dhillon - Manag Econ Res J, 2019 - merj.scholasticahq.com To comprehend Sino–US trade relations, this research article decrypts the trade relations among China and the United States from the American government perspective (Presidency of Donald Trump). The American government claims that the Chinese government's high ... $\Delta \Sigma \Sigma$ Related articles $\delta \delta$

China US Trade War: Will it impact on Latin American Countries?

A Peluffo - Serie Documentos de Trabajo; 08/20, 2020 - colibri.udelar.edu.uy This article gives a brief overview of the current trade war between China and the US, pointing out its main but multiple causes that lead to the current trade war. Then we take a look to the likely effects on the contenders. Finally, we analyze briefly the probable effects ... $\frac{1}{20}$ All 4 versions \gg

Optimum Renewable Energy Investment Planning in Terms of Current Deficit: Turkey Model

SY Saçık, N Yokuş, M Alagöz, T Yokuş - Energies, 2020 - mdpi.com In this study, a methodology was suggested for wind and solar energy investment plans through linear optimization model for the countries with an energy-based current deficit problem. The originality of the study is that it is a renewable energy investment model based ... ☆ ワワ Cited by 1 Related articles All 9 versions 🎊

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

Trade Wars and the Disarray in the Global Trading System: Implications for the Philippines

MJV Abrenica, RRS Guzman... - Asian Economic ..., 2019 - MIT Press

This study uses the Caliendo and Parro multi-sector, multi-country, general equilibrium Ricardian trade model with national and international input-output linkages to assess the impact on welfare of higher tariffs due to the US–China trade war in the case of the ... $\frac{1}{29}$ Related articles All 4 versions

China-US trade: implications on conflicts

LD Qiu, X Wei - China Economic Journal, 2019 - Taylor & Francis

In this study, we examine various aspects of China's trade, the US'trade, and the bilateral trade between the two countries. The analysis of each aspect has direct and indirect implications on trade conflicts between the two countries. We focus on important factors ...

 $\cancel{2}$ 99 Cited by 1 Related articles All 3 versions

Reordering international trade: what will it cost?

R Stasiukynaitė - 2019 - zbw.eu

This paper overviews early research which has gone into the possible effects of the recent tensions in international trade. We have witnessed the increase and the promises of future increase in tariffs, in the case of the new trade policy of the US, and the promises of the ... $\sum_{n=1}^{\infty}$ $\frac{99}{20}$ Related articles All 5 versions \gg

China-US Trade War and Trade Talk

M Yu - Springer Books, 2020 - Springer

China's opening-up in the past four decades has gone through three waves: the extensive margin of opening-up (1978–2001), the intensive margin of opening-up (2001–2017), and all-around opening-up (since 2017). This paper explores these three stages of the country's ... $\sum_{n=1}^{\infty}$ $\frac{99}{n}$ Related articles All 6 versions

How Resilient Is ASEAN-5 to Trade Shocks? A Comparison of Regional and Global Shocks

M Raghavan, ES Devadason - Global Journal of Emerging ..., 2020 - journals.sagepub.com This article studies the resilience of the ASEAN region to external shocks amid the unfolding effects of the USA–China trade war. It investigates and compares the effects of regional (ASEAN) and global (USA, China) shocks on ASEAN-5 using a Structural VAR (SVAR) ... \therefore Σ Σ Cited by 1 Related articles All 4 versions

The Status of China's Market Economy and Structural Reforms: The Issues

M Yu - 2019 - en.ccer.pku.edu.cn

1 I am indebted to Wing-Thye Woo for his warm invitation and insightful advice and suggestions on this project. I also thank two anonymous referees, Ming Lu, Dwight Perkins, Deborah Swenson, and participants of the Asian Economic Panel Meeting in Kuala Lumpur ... ☆ 99 Related articles All 8 versions ≫

Who will win from the trade war? Analysis of the US–China trade war from a micro perspective

V Archana - China Economic Journal, 2020 - Taylor & Francis

Over the past two years, China and the US are going on a trade war imposing tariffs on one another's goods and this is hurting each other's businesses. The focus of the present paper is to analyse the costs and benefits of tariff policy changes by the two economies. A partial ... \therefore 55

Nuclear Fusion: The Case for a High-Level Global External Independent Review

J Draper - 2019 - osf.io

Fusion energy has the potential to address long-term energy requirements and, again over the long term, climate change. However, the development of fusion energy is characterized by a 'development divide'between International Thermonuclear Experimental Reactor (ITER) ... $\frac{1}{22}$ Related articles All 4 versions \gg

China and the US-The Behemoths Trade War

M Chen - 2020 - repository.usfca.edu

China and the US are the world's largest economies and the largest mutual trade partners. The US has also held the position of the world's largest economy for most of the current fiscal age. However, China is rising, largely bolstered by a great population, resource ... $\frac{1}{\sqrt{29}}$ Related articles $\frac{1}{\sqrt{29}}$

USF Scholarship: a digital repository@ Gleeson Library| Geschk e Cente r

M Chen - 2020 - core.ac.uk

China and the US are the world's largest economies and the largest mutual trade partners. The US has also held the position of the world's largest economy for most of the current fiscal age. However, China is rising, largely bolstered by a great population, resource ...

☆ 99 Related articles ≫

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

How does "the Belt and Road" and the Sino-US trade conflict affect global and Chinese CO 2 emissions?

JL Fan, Y Dong, X Zhang - Environmental Science and Pollution Research, 2020 - Springer In the context of the rapid development of the Belt and Road (B&R) Initiative, the continuous transfer of Sino-US trade to the B&R countries is an important means to mitigate the threat of Sino-US trade, and the environmental impact of this transfer should be considered, so as to ... $\frac{1}{20}$ All 3 versions

Time Series Analyses of Global Oil Prices: Shocks, Effects and Predictability

S Ruths Sion - 2020 - tuprints.ulb.tu-darmstadt.de

This dissertation considers different aspects of crude oil research, primarily based on four independent empirical analyses, interconnected through a common denominator: Time-series analysis methods applied to global oil prices. The first three chapters are of ...

☆ ワワ All 5 versions ≫

Environmental and economic impacts of trade barriers: The example of China–US trade friction

LJ Liu, F Creutzig, YF Yao, YM Wei... - Resource and Energy ..., 2020 - Elsevier The new age of trade wars could simultaneously affect the worldwide distribution pattern of the economy and environmental emissions. However, previous studies have focused on economic impacts, and on trade liberalization, while little is known about the equilibrium ...

 $\cancel{2}$ $\cancel{99}$ Cited by 2 Related articles All 6 versions

" Economic Consequences of The Current US Trade Policy

R Dos Santos, G Vannoorenberghe - dial.uclouvain.be

This thesis analyzes current United States Trade Policy in order to review and understand recent events in the world trade system. The content of the 2018 US Trade Policy Agenda having led to greater uncertainty, this analysis shows different possible future scenarios in ... \therefore \Im Related articles All 2 versions \gg

International competition and trade conflict in a dual photovoltaic supply chain system

Z Chen, SII Su - Renewable Energy, 2020 - Elsevier

This study has explored a contemporary trade conflict issue heated up by the trade war between two largest economies in the world. Since the solar industry is a fast developing and critical renewable energy industry sector with many global suppliers competing fiercely ...

 $\cancel{2}$ 99 Cited by 1 Related articles All 7 versions

Two Conceptions of Global Order

L Zucca - A Legal Analysis of the Belt and Road Initiative, 2020 - Springer Trump's America is dismantling the global order the United States helped to create. Meanwhile, China is emerging as the leading economic power. To affirm itself, China will

have to rely on international institutions that facilitate trade and exchange with the rest of the ...

 $\cancel{2}$ $\cancel{99}$ Related articles All 6 versions

The Trump administration's politics of time: The temporal dynamics that enable Trump's interests to determine American foreign policy

C McIntosh - Time & Society, 2020 - journals.sagepub.com Scholars and political observers have noted how the rate of significant events and crises in the Trump administration appear to be accelerating at an unmanageable and irresistible pace, making it difficult to respond to these actions as they happen. This paper uses the lens ...

 $\cancel{2}$ $\cancel{99}$ Related articles

Yu Miaojie: Evaluating the Burden of a US-China Trade War

M Guo, L Lu, L Sheng, M Yu - en.nsd.pku.edu.cn

Trade disputes between the United States and China greatly intensified recently as the two countries announced a 25 percent tariff hike on \$50 billion worth of products imported from each other, raising the risk of a trade war between the two giant trading economies. Based ... $\sum_{n=1}^{\infty}$ $\overline{99}$ Related articles All 2 versions \gg

Competition strategy and governance policy for the dual international supply chains under trade war

Z Chen, J Peng - ... Journal of Management Science and Engineering ..., 2020 - Taylor & Francis Against the backdrop of the full-blown trade war, this study investigates the price discrimination strategies and tariff policies for the dual international supply chains in trade war via game-theoretical modeling and numerical and sensitivity analyses approaches ... $\frac{1}{\sqrt{2}}$ $\frac{99}{20}$ Related articles All 2 versions

The Korea-Japan trade dispute: non-tariff barriers

S Shin - 2020 - econ.iastate.edu

In mid-2019, a new major trade war started to heat up in east Asia, while the US-China trade war held the spotlight. This article documents the recent Korea-Japan trade dispute and quantifies economic impacts. We calibrate the GTAP version 10 database to the ...

☆ ワワ ≫

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

Will Increasing Tariffs on China Really Bring the Manufacturing Plants Back to the US?

L Wang, GJD Hewings - Global Economic Review, 2020 - Taylor & Francis This paper investigates whether the recent rise in tariffs on goods produced in China will lead processing trade manufacturing plants now located in China to delocate to the US By using a hypothetical extraction method and examining the global value chains of income, we ... 22 Related articles All 7 versions 5

Trade war and the economy

L Cheah, SJ Koh, YY Lim - 2019 - eprints.utar.edu.my

Trade war is mostly known as the economic conflicts between two and or more countries that impose tariff to harm each other's trade. Trade war are no longer a new phenomenon in this century and are often known to economists and individuals. In 2018, United States triggered ... ☆ 22 Related articles 88

TRADE COMPETITION BETWEEN USA AND MALAYSIA UNDER THE SHADOW OF FREE TRADE

K Osman

22 ☆ **Related articles**

THE IMPACT OF SINO-USA TRADE WAR ON USA FARM BANKRUPTCIES

Y Wu - 2020 - ecommons.cornell.edu

The Sino-USA trade war has raised concerns on the USA agricultural economy as farm bankruptcies rose up. While media reports tend to attribute the increase in farm bankruptcy to the trade war, there are many other factors affecting the farm sector. Therefore, the ... 22 88 ŵ

Why is the Hong Kong housing market unaffordable? Some stylized facts and estimations

CK Leung, CYJ Ng, E Tang - Some Stylized Facts and Estimations ..., 2020 - papers.ssrn.com The house price in Hong Kong is well-known to be" unaffordable." This paper argues that the commonly used house price-to-income ratio may be misleading in an economy with almost half of the population living in either public rental housing or subsidized ownership ... ☆ 22 Related articles All 11 versions

Let's talk... about farmers: Trump's Trade War and a looming mental health crisis.

DJ Herman - University of Toronto Medical Journal, 2019 - search.ebscohost.com Donald Trump's aggressive stance on trade has led to heavy tariffs on US agricultural products. This has exposed American farmers to a high amount of volatility and work stress. Suicides in America have drastically increased since the late 1990's, and agriculture has ...

22 **Related articles** ক্স

Working Papers in Trade and Development

P Athukorala - 2009 - crawford.anu.edu.au

This paper examines the implications of global production sharing for economic integration in East Asia, with emphasis on the behaviour of trade flows in the wake of the 2008 global economic crisis. It is found that, while 'network trade'has generally grown faster than total ☆ 99 SS

Related articles All 29 versions

Implications of the US-China Trade War for Taiwan

W Hsieh - Asian Economic Papers, 2020 - MIT Press The ongoing US-China trade war and ensuing high-tech conflicts are regarded as Taiwan's most crucial opportunity to slow down its progressively increasing economic dependence on China. The impact of the US-China trade tensions on Taiwan are important to analyze ...

22 Related articles All 5 versions ক্স

全球价值链研究综述及前景展望

杨翠红, 田开兰, 高翔, 张俊荣 - 系统工程理论与实践, 2020 - sysengi.com

The improvement of information and communication technology and the reduction in trade costs promoted the formation of global value chains (GVCs). The expansion of GVCs further drove the evolution of international trade theories and the updates of trade accounting ...

☆ ワワ All 5 versions ≫

Diplomacia comercial China-Estados Unidos: enfoques metodológicos y sistémicos

JI Lechuga-Cardozo - Razón Crítica, 2020 - revistas.utadeo.edu.co El presente artículo es producto de una revisión bibliográfica, cuyo objetivo consistió en analizar los diferentes enfoques internacionales que explican la diplomacia comercial actual entre China y Estados Unidos desde las relaciones internacionales. Se aplicó el ...

☆ ワワ Related articles ≫

The day after tomorrow: Evaluating the burden of Trump's trade war

Search within citing articles

中美贸易战:基于动态竞争的视角

陈明哲, 庞大龙 - 外国经济与管理, 2019 - qks.sufe.edu.cn 2018年,中美贸易战由美国最高元首特朗普亲自发动,对中国,美国乃至世界均产生了深远的 影响. 一些文章已经对贸易战相关议题发表看法, 然而既有文献对中美双方在贸易战演进过程中, 竞争行动的交互性, 动态性及多领域互动的本质仍然缺乏系统的探究. 因此 ... 99 Related articles All 4 versions £₹ 8

Unión Europea y Japón: ¿ El tratado de libre comercio más grande del mundo?

JJB Rojas, GC Montoya - Revista de ciencias sociales, 2019 - dialnet.unirioja.es Durante más de un lustro la Unión Europea y Japón han venido llevando a cabo una compleja negociación para alcanzar la firma de un Tratado de Libre Comercio que reduzca en general las barreras arancelarias e igualmente las no arancelarias a razón de liberalizar ... ☆ 22 Cited by 2 Related articles All 4 versions 88

Gediminas VALANTIEJUS

S MOKSLAI - researchgate.net

Tyrimo aktualumas ir problematika. Muitų išskirtinę reikšmę tarptautinės prekybos reguliavimui bei jos liberalizavimui ir su tuo susijusių teisinių priemonių tinkamo taikymo svarbą pažymi daugelis šiuolaikinę tarptautinės prekybos ir Pasaulio Prekybos ... 55

99 Related articles ☆

Muitų teisinio reguliavimo ypatumai tarptautinėje prekyboje tarp Europos Sąjungos ir Kinijos Liaudies Respublikos, Rusijos Federacijos bei Indijos Respublikos

G Valantiejus - 2019 - repository.mruni.eu

Tyrimo aktualumas ir problematika. Viena iš pagrindinių šiuolaikinės, XXI a. ekonominės pasaulio sąrangos ypatybių, tiesiogiai įtakojančių tiek tarptautinės, tiek ir nacionalinių teisinių sistemų funkcionavimą, yra globalumas (globalios ekonomikos sistemos ... 55

22 Related articles All 3 versions ☆

ГОСПОДАРСЬКО-ПРАВОВІ ЗАСОБИ ЗАБЕЗПЕЧЕННЯ ЕКОНОМІЧНОЇ БЕЗПЕКИ УКРАЇНИ

ЄМ БІЛОУСОВ - Національний університет імені Ярослава ..., 2018 - nauka.nlu.edu.ua Дисертація на здобуття наукового ступеня доктора юридичних наук за спеціальністю 12.00. 04 «Господарське право; господарськопроцесуальне право».-Національний юридичний університет імені Ярослава Мудрого, Міністерство освіти і науки України ...

☆ 22 Cited by 1 Related articles All 2 versions 88